Author's response to reviews

Title: CT Colonography Imaging of Pneumatosis Intestinalis after Hyperbaric Oxygenotherapy

Authors:

jean louis frossard (jean-louis.frossard@hcuge.ch)
philippe braude (philippe.braude@colline.ch)
jean yves berney (jean-yves-berney@hcuge.ch)

Version: 4 Date: 1 March 2011

Author's response to reviews: see over
To The Editors

Hôpital Cantonal Universitaire de Genève  
Division de Gastroentérologie

Genève, le 1 mars 2011

N/Réf. :

RE : submission of a second revised version of our manuscript

Dear Editor,

Enclosed you will find our revised manuscript entitled “CT Colonography Imaging of Pneumatosis Intestinalis after Hyperbaric Oxygenotherapy” that we would like to be considered for publication in your Journal. Hereafter you will find our replies to the reviewers’s queries.

Point by point replies to the reviewers:

Reviewer

Ethnicity has been added as suggested in the case presentation. Age, sex ethnicity have been also added in the abstract section.

This is true that Mr Nicolazzi is the patient. He gave his formal acceptance for being included as an author since he is the key point in this case. He made a substantial contribution as a patient by having several MRI. However, if the author list cannot contain his name because he does not qualify as an author, we would like to acknowledge him at the end of the manuscript.

A conclusion section has been added as suggested and now conforms to the journal style.

I thank you in advance for your consideration

Sincerely yours

Jean Louis Frossard, MD